Author's response to reviews

Title: Pressure ulcer incidence in Dutch and German nursing homes: design of a prospective multicenter cohort study

Authors:

Esther Meesterberends (e.meesterberends@maastrichtuniversity.nl)
Ruud JG Halfens (r.halfens@maastrichtuniversity.nl)
Cornelia Heinze (cornelia.heinze@charite.de)
Christa Lohrmann (christa.lohrmann@medunigraz.at)
Jos MGA Schols (jos.schols@maastrichtuniversity.nl)

Version: 2 Date: 18 April 2011

Author's response to reviews: see over
Dear Dr. Patel,

Hereby we would like to submit the revision of the manuscript entitled “Pressure ulcer incidence in Dutch and German nursing homes: design of a prospective multicenter cohort study” to BMC nursing.

Based on the requested formatting changes we have made the following revisions to the manuscript:

1. We have changed the table in our manuscript according to the changes requested.
2. The order on the title page has been changed.
3. Countries have been included in the affiliation details
4. We have checked the manuscript for typographical errors and we have made final corrections to the manuscript

We genuinely hope that the manuscript is suitable for publication in this revised form.

Yours sincerely,

Esther Meesterberends, Ruud Halfens, Cornelia Heinze, Christa Lohrmann and Jos Schols

Esther Meesterberends, PhD student, RN, MSc
Faculty of Health, Medicine and Life Sciences
School for Public Health and Primary Care (CAPHRI)
Department of Health Care and Nursing Science
P.O. Box 616, 6200 MD Maastricht, The Netherlands
Tel. +31 43 3881553 Fax +31 43 3884162
Email: e.meesterberends@maastrichtuniversity.nl